Author’s response to reviews

Title: Early-onset ventilator-associated pneumonia incidence in intensive care units: a surveillance-based study

Authors:

Philippe Vanhems (philippe.vanhems@chu-lyon.fr)
Thomas Bénet (thomas.benet@chu-lyon.fr)
Nicolas Voirin (nicolas.voirin@chu-lyon.fr)
Jean-Marie Januel (jean-marie.januel@chuv.ch)
Alain Lepape (alain.lepape@chu-lyon.fr)
Bernard Allaouchiche (bernard.allaouchiche@chu-lyon.fr)
Laurent Argaud (laurent.argaud@chu-lyon.fr)
Dominique Chassard (dominique.chassard@chu-lyon.fr)
Claude Guerin (claude.guerin@chu-lyon.fr)

Version: 3 Date: 25 July 2011

Author’s response to reviews:

Dear Sir/Madam,

Please find enclosed the revised version of our manuscript entitled “Early-onset ventilator-associated pneumonia incidence in intensive care units: a surveillance-based study” for submission as research article.

The review was done by a native-English person with scientific expertise.

If you think that additional editing is needing, we are ready to improve again the text.

Yours sincerely,

Philippe Vanhems, MD, PhD